CSCI 1323 (Discrete Structures), Spring 2001 Homework 17

Assigned: March 29, 2001. Due: April 5, 2001. Credit: Not for credit.

1 Problems

1. Do problem 11 on p. 194 of the textbook. 2. Do problem 21 on p. 195 of the textbook. 3. Do problem 26 on p. 196 of the textbook. 4. Do problem 59 on p. 197 of the textbook. 5. Do problem 8 on p. 203 of the textbook. 6. Do problem 14 on p. 204 of the textbook. 7. Do problem 5 on p. 215 of the textbook. 8. Do problem 10 on p. 215 of the textbook. 9. Do problem 18 on p. 216 of the textbook. 10. Do problem 23 on p. 216 of the textbook. 11. Do problem 44 on p. 217 of the textbook. 12. Do problem 49 on p. 217 of the textbook. 13. Do problem 55 on p. 218 of the textbook. 14. Do problem 59 on p. 218 of the textbook. 15. Do problem 67 on p. 219 of the textbook. 16. Do problem 73 on p. 220 of the textbook.